Author's response to reviews

Title: A systematic review on the effect of sweeteners on glycemic response and clinically relevant outcomes

Authors:

Natasha Wiebe (nwiebe@ualberta.ca)
Raj Padwal (rpadwal@ualberta.ca)
Catherine Field (catherine.field@ualberta.ca)
Seth Marks (smarks@ualberta.ca)
Rene Jacobs (rjacobs@ualberta.ca)
Marcello Tonelli (mtonelli@ualberta.ca)

Version: 3 Date: 29 August 2011

Author's response to reviews: see over
29 August, 2011

Dear BMC Series Editorial Production Team:

Re: MS: 1920640989562119
A systematic review on the effect of sweeteners on glycemic response and clinically relevant outcomes
Natasha Wiebe, Raj Padwal, Catherine Field, Seth Marks, Rene Jacobs and Marcello Tonelli

Thank you for your comments. I have revised the proof accordingly. If you have further questions, or require further information, please don’t hesitate to contact me.

Sincerely,
Natasha Wiebe for the authors
Emails: Please ensure that all authors email addresses on the title page of the manuscript are identical to those in the submissions system.
Done.

References: please provide a functioning links to the websites in references 2 and 20. I clicked both references and the pdfs popped up so I am not sure why you cannot access these pdfs.

References cannot be cited under the tables as they need to be cited within the main text within the main reference list. Please rectify this and include references to your table by having numerical references in square brackets. Like the rest of the text, please ensure all references are cited in numerical order throughout the text and tables.
Done.

Authors' contributions: Please indicate that all authors read and approved the final manuscript.
Done.

Conclusions: the information contained within the Main Findings box must be incorporated into the main text body or moved to the conclusions section as it cannot come at the very end of the manuscript.
I moved the box to follow the paragraph that mentions it.

Additional files: all additional files must be mentioned in the text in numerical order, or removed from the system
There is only 1 additional file.

Figures: Please provide a description of no more than 300 words of what Figure 1 demonstrates in the Figure Legends section.
Done.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
No further changes to make.